Author's response to reviews

Title: Impact of Introducing Multiple Evidence-Based Clinical Practice Protocols in a Medical Intensive Care unit: A Retrospective Cohort Study

Authors:

Bekele Afessa (afessa.bekele@mayo.edu)
Ognjen Gajic (gajic.ognjen@mayo.edu)
Mark T Keegan (gajic.ognjen@mayo.edu)
Edward G Seferian (seferian.edward@mayo.edu)
Rolf D Hubmayr (rhubmayr@mayo.edu)
Steve G Peters (peters.steve@mayo.edu)

Version: 4 Date: 1 August 2007

Author's response to reviews:

Dear editor,

As per the requests for appropriate formatting, we have revised the manuscript.

1. We have reformatted the competing interest section stating that "The authors declare that they have no competing interests."

2. The funding information was not a competing interest. We have moved it to the acknowledgment section and stated that the funding Office of Faculty Development played no role in any aspects of the study.

3. We have reformatted "Results" and "Discussion"

4. We have added the country name (USA) to each of the affiliations.

Sincerely,

Bekele Afessa, MD